Response to Editorial request.

1 - Please include the names of the members of Australian Ovarian Cancer Study Group in the Acknowledgements section of the manuscript.

The following information has been added to the Acknowledgements section:

Hammond, A McCartney (dec.), C Stewart, Y Leung, M Buck, N Zeps (WARTN). Further information can be found at http://www.aocstudy.org.

2 - Please ensure that any sample identifiers in the tables are not traceable to any of the participants in the study.

We confirm that the sample identifiers cannot be traced to study participants.